%% To students who wish to submit their homework in LaTeX format:
%% On the line that starts ``\newcommand{\student}''
%%   put your name(s) in the appropriate place.
%% Then, uncomment the line further down that starts
%%   ``%% \\[2pt]\student''

\documentclass[11pt]{article}

\usepackage{fullpage}

\newcommand{\solution}{\bigskip\hrule\bigskip}
\newcommand{\problembreak}{\bigskip\hrule\bigskip}

\renewcommand{\theenumi}{\bf\Alph{enumi}}

\newlength{\spread}
\setlength{\spread}{1.5in}

\begin{document}

\thispagestyle{empty}

\begin{center}
{\Large
CS5114 Spring 2014\\
Solution to Homework Assignment \PutAssignmentNumberHere
}
\end{center}
\vspace*{0.10in}
\medskip

\textbf{Pledge: I (we) understand the answers that I (we) have
submitted.
The answers submitted have not been directly copied from another
source, but instead are written in my (our) own words.}

\bigskip
\PutYourNameHere, \PutYourIDNumberHere

%Put second name/ID here if there is one

\bigskip
\textbf{1.} \PutProblemNumberHere

\solution

% PUT YOUR SOLUTION HERE
% WARNING: IF YOU WORK WITH A PARTNER, BE SURE TO INCLUDE A STATEMENT
% THAT INDICATES THE CONTRIBUTIONS OF EACH PERSON!

\problembreak

\bigskip
\textbf{2.} \PutProblemNumberHere

\solution

% PUT YOUR SOLUTION HERE
% WARNING: IF YOU WORK WITH A PARTNER, BE SURE TO INCLUDE A STATEMENT
% THAT INDICATES THE CONTRIBUTIONS OF EACH PERSON!

\problembreak

\bigskip
\textbf{3.} \PutProblemNumberHere

\solution

% PUT YOUR SOLUTION HERE
% WARNING: IF YOU WORK WITH A PARTNER, BE SURE TO INCLUDE A STATEMENT
% THAT INDICATES THE CONTRIBUTIONS OF EACH PERSON!

\problembreak

\end{document}
